Author's response to reviews

Title: Financial Impact of adopting Implantable Loop Recorder Diagnostic for unexplained syncope compared with Conventional Diagnostic Pathway in Portugal

Authors:

Rui Providencia (rui_providencia@yahoo.com)
Rui Candeias (candeias_rui@hotmail.com)
Carlos Morais (moraiscms@gmail.com)
Hipólito Reis (a.hipolitoreis@gmail.com)
Luis Elvas (luisdvelvas@netcabo.pt)
Vitor Sanfins (vsanfins@netcabo.pt)
Sara Farinha (sara.p.farinha@gmail.com)
Simon Eggington (simon.eggington@medtronic.com)
Stelios Tsintzos (stelios.tsintzos@medtronic.com)

Version: 3 Date: 11 September 2013

Author's response to reviews:

The following sentences were added at the end of the methods section, according to what was suggested: “No ethical approval was required for this study as it consisted on a secondary analysis of previously published data. Data analyzed in this study was publicly available and all the used sources are adequately cited throughout the manuscript.”